Author’s response to reviews

Title: Professional centred shared decision making: patient decision aids in practice in primary care

Authors:

Duika Burges Watson (duika.burges-watson@ncl.ac.uk)
Richard G Thomson (richard.thomson@ncl.ac.uk)
Madeleine J Murtagh (m.j.murtagh@newcastle.ac.uk)

Version: 6 Date: 21 November 2007

Author’s response to reviews:

We have responded to all the minor essential revisions as requested by reviewer 2, Frances Legare. That is we have

1. Moved the text `our approach differs¿ From the introduction to the discussion
2. Edited all references
3. Provided the BMJ reference for IPDAS
4. Removed `shared decision making¿ and replaced with SDM in the body of the article
5. Provided clarification of `FG1¿ etc
6. Moved ¿ Our findings suggest¿¿ from results to the first paragraph of the discussion session
7. Added the reference for Elwyn